Author's response to reviews

Title: Hertfordshire sarcopenia study :design and methods

Authors:

Harnish P Patel (hp@mrc.soton.ac.uk)
Holly E Syddall (hes@mrc.soton.ac.uk)
Helen J Martin (hjm@mrc.soton.ac.uk)
Claire E Stewart (c.stewart@mmu.ac.uk)
Cyrus Cooper (cc@mrc.soton.ac.uk)
Avan Aihie Sayer (aas@mrc.soton.ac.uk)

Version: 3 Date: 25 June 2010

Author's response to reviews: see over
Thank you for publishing our study protocol. We have revised it as instructed.

1. Title now in sentence case. Running title has been removed and text highlighting (yellow as well as bold) has been removed from the manuscript. The email list has been simplified.

2. Key words have been removed.

3. Placement text have been removed.

4. All cited references are now in numerical order. The references now contain all authors.

5. Table and figure legends have been moved to after the references.

6. Figure 1 has been redrawn and cropped to minimize white space. Figure 3 has been removed completely.